Author’s response to reviews

Title: Repetition of suicide attempts across episodes of severe depression
Behavioural sensitisation found in suicide group but not in controls

Authors:

Louise Brådvik (louise@bradvik.se)
Mats Berglund (Mats.Berglund@med.lu.se)

Version: 5 Date: 4 January 2011

Author’s response to reviews:

Point-by-point revision of MS ID : 2032317877465071
Repetition of suicide attempts across episodes of severe depression
Behavioural sensitisation found in suicide group but not in controls

P7 line 8 “committed suicide” has been changed to “died by suicide”, as suggested.

Furthermore, we have received more up-to-date death statistics and the entire sample is now follow up to 2010-02-15. Therefore 2006 (previous follow up) has been changed to 2010 all through the paper.